Author's response to reviews

Title: Perceptions of the Critical Cultural Competence of Registered Nurses in Canada

Authors:

Adel Almutairi (almutairiadj @ngha.med.sa)
Abdullah Adlan (adlanA @ngha.med.sa)
Maliha Nasim (nasimma@NGHA.MED.SA)

Version: 1 Date: 28 May 2017

Author’s response to reviews:

Dear reviewers,

We appreciate the thoughtful reviews and helpful comments on our manuscript. In the attached letter we explain how we have attended to these comments. We believe that the paper is much stronger now. Changes that have been made in response to reviewers’ feedback are indicated in the revised manuscript in red font.

The attached letter also contains a table that itemizes the reviews’ comments and our corresponding responses. Thank you very much for the opportunity to resubmit a revised manuscript.